Author's response to reviews

Title: Utility of 18F-fluoro-deoxyglucose emission tomography / computed tomography fusion imaging (18F-FDG PET/CT) in combination with ultrasonography for axillary staging in primary breast cancer.

Authors:

Shigeto Ueda (syueda2000@yahoo.co.jp)
Hitoshi Tsuda (htsuda@ndmc.ac.jp)
Hideki Asakawa (hiddle_vr4@hotmail.com)
Jiro Omata (dr20008@ndmc.ac.jp)
Kazuhiko Fukatsu (kfukatsu@ndmc.ac.jp)
Nobuo Kondo (nkondo@ndmc.ac.jp)
Tadaharu Kondo (kondo@ndmc.ac.jp)
Yukihiro Hama (yjhama2005@yahoo.co.jp)
Katsumi Tamura (abe@toko-pet.or.jp)
Jiro Ishida (abe@toko-pet.or.jp)
Yoshiyuki Abe (abe@toko-pet.or.jp)
Hidetaka Mochizuki (mochizuki@ndmc.ac.jp)

Version: 7 Date: 30 May 2008

Author's response to reviews: see over
May 30, 2008

Editor-in-Chief Melissa Norton, MD

Dear Dr. Melissa Norton,

RE: No.

Title: Utility of $^{18}$F-fluoro-deoxyglucose emission tomography / computed tomography fusion imaging in combination with ultrasonography for axillary staging in primary breast cancer.

Thank you very much for your E-mail. We have confirmed no revise of the contents of our manuscript. We appreciate your help for publication.

Sincerely yours,

Hitoshi Tsuda, M.D.
Department of Basic Pathology, National Defense Medical College
Namiki 3-2, Tokorozawa, Saitama, Japan
e-mail: htsuda@ndmc.ac.jp
TEL. +81-4-2995-1211
FAX. +81-4-2996-5193